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ABSTRACT 

 

Single step methane conversion to value added products is an economical route to exploit 

natural gas reservoirs from industrial perspective. Conversion of methane to aromatics in non-

oxidative conditions generally utilizes a bifunctional catalyst having active molybdenum 

species supported on zeolite (Mo/HZSM-5). In this reaction, the nature of the active site for 

methane on Mo is suggested to be a carbide (or an oxycarbide) and the catalyst is known to 

deactivate from carbon deposition over time due to the formation of polyaromatic 

hydrocarbons. The evolution of the molybdenum carbide species from the precursor oxide and 

oxycarbide forms, dictating the anchoring of the carbide cluster on the zeolite support, is 

considered the key aspect in deciding the extent of coke formation on the catalyst. Herein, 

density functional theory (DFT) calculations are utilized to study the catalytically active 

carbide (MoxCy) clusters in C-H bond activation and C-C coupling reaction to form C2 

intermediates. A massively parallel cascade genetic algorithm (cGA) is applied to scan the 

potential energy surface (PES) for all possible low-energy structures (including the global 

minimum) of MoxCy. Catalytic activity of the global minimum and metastable clusters are 

assessed as well as compared for methane activation, highlighting the role of metastable 

clusters influencing the overall reaction rates. The mechanistic insights obtained from DFT 

calculations thus provides a molecular level engineering approach wherein, the catalyst is 

rationally synthesized to obtain desired active sites anchored on the zeolite support. These 

active sites are stable and responsible for providing high aromatic selectivity and the overall 

reactivity. This is implemented in experiments at three stages; a) this is achieved by obtaining 

a higher amount of Mo carbide by tailoring the carburizing agent. Changing the carburizing 

condition from pure methane to a mixture of hydrocarbon gases (mimicking the natural gas). 
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In this reference, we studied the effect of mixture of C2H6, C3H8, C4H10, and methane along 

with hydrogen on molybdenum carbide formation during the initial carburization step; b) 

another strategy to stabilize these primary molybdenum oxide/carbide species over the zeolite 

after subsequent thermal and chemical treatment is proposed. It is achieved by thermal 

treatment of molybdenum precursors and conversion of orthorhombic α-MoO3 to metastable 

hexagonal h-MoO3 via hydrothermal topotactic transformation. In this work, the role of Mo 

precursor interaction with the zeolite in altering catalyst reactivity is studied; and c) in order to 

achieve higher aromatic (benzene) selectivity the brØnsted acidity of HZSM-5 zeolite is 

moderated by pretreating it with boric acid. Moreover, this novel strategy of controlling the 

brØnsted acidity of HZSM-5 zeolite enhances catalyst stability in MDA by reducing the 

agglomeration of active Mo2C6 species and the carbon deposition. Overall, through these 

studies, the dissertation provides molecular level insights of the active catalyst sites of the 

reaction. Nonetheless, this study provides a guiding tool for rational catalyst design by 

specifying the choice of Mo precursor, carburizing conditions and zeolite treatment for 

obtaining desired reactivity and product selectivity, with a potential to achieve higher catalyst 

stability for MDA process.  
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सार 

औद्योगिक दृष्टिकोण से  प्राकृष्टिक गैस के भंडार को मीथेन के एकल चरण मे मूल्य वगधित उत्पादएस मे 

परिवतिन करना एक ष्टकफायिी मागग ह । गैर-ऑक्सीडेष्टिव स्थिष्टियो ंमें मीिेन का एरोमेष्टिक्स में रूपांिरण 

आम िौर पर ष्टिओलाइि (Mo/HZSM-5) पर स्थिि ष्टि-कायागत्मक उते्प्ररक मौिूद सष्टिय मोष्टलब्डेनम 

थिलो ंका उपयोग करिा है। यह माना िािा ह की इस प्रष्टिष्टिया में, मोष्टलब्डेनम पर मीिेन के ष्टलए 

सष्टिय थिलो ंकी प्रकृष्टि को कार्ागइड (या एक ऑक्सीकार्ागइड) है और समय के साि उते्प्ररक की सतह 

पर पॉलीएरोमैष्टिक हाइडर ोकार्गन के गठन के कारण कार्गन िमाव से उते्प्ररक ष्टनस्िय हो िािा है। 

मोष्टलब्डेनम कार्ागइड अणुओ ंका ऑक्साइड और ऑक्सीकार्ागइड के रूपो ंसे ष्टवकास, ष्टिओलाइि नी ींव 

पर कार्ागइड क्लस्टर के लंगर को ष्टनर्ागररि करिा  है औि यह माना जाता है  की उते्प्ररक पर कोक गठन 

की सीमा िय करने में महत्वपूणग पहलू है। इस प्रकार, घनत्व कायागत्मक ष्टसद्ांि (DFT) गणनाओ ंका 

उपयोग C-H रं्र्न और C-C युग्मन प्रष्टिष्टिया से C2 मध्यविी र्नाने के ष्टलए  उते्प्ररक सष्टिय कार्ागइड 

(MoxCy) समूहो ंका अध्ययन ष्टकया िया है। MoxCy की सभी संभाष्टवि कम-ऊिाग संरचनाओ ं (वैष्टिक 

नू्यनिम सष्टहि) के ष्टलए संभाष्टवि ऊिाग सिह (PES) की जााँच करने के ष्टलए एक व्यापक समानांिर 

कैसे्कड आनुवंष्टिक एल्गोररथ्म (cGA) इसिेमाल ष्टकया िािा है। वैष्टिक नू्यनिम और मेिासे्टर्ल समूहो ं

की उते्प्ररक गष्टिष्टवष्टर् का मूल्ांकन ष्टकया िािा है और साि ही मीिेन सष्टियण की िुलना में, समग्र 

प्रष्टिष्टिया दरो ंको प्रभाष्टवि करने वाले मेिासे्टर्ल समूहो ंकी भूष्टमका पर प्रकाि डाला गया है। डीएफिी 

गणनाओ ंसे प्राप्त यांष्टिकी अंिदृगष्टि इस प्रकार एक आणष्टवक स्तर इंिीष्टनयररंग दृष्टिकोण प्रदान करिी 

है, ष्टिसमें ष्टिओलाइि समिगन पर लगी वांष्टिि सष्टिय साइिो ंको प्राप्त करने के ष्टलए उते्प्ररक को 

िकग संगि रूप से संशे्लष्टिि ष्टकया िािा है। ये सष्टिय साइिें एरोमेष्टिक्स चयनात्मकिा और समग्र 

प्रष्टिष्टियािीलिा प्रदान करने के ष्टलए स्थिर और ष्टिमे्मदार हैं। इसे िीन चरणो ंमें प्रयोगो ंमें लागू ष्टकया 



 

viii 
 

गया है; क) यह कार्गराइष्टिंग प्रष्टिष्टनष्टर् का अनुरूपण करके अष्टर्क मािा में मोष्टलब्डेनम कार्ागइड प्राप्त 

करके प्राप्त ष्टकया िािा है। िुद् मीिेन से कार्ोराइष्टिंग स्थिष्टि को हाइडर ोकार्गन गैसो ंके ष्टमश्रण में 

र्दलना (प्राकृष्टिक गैस की नकल करना)। इस संदभग में, हमने प्रारंष्टभक कार्गराइिेिन चरण के दौरान 

मोष्टलब्डेनम कार्ागइड के गठन पर हाइडर ोिन के साि C2H6, C3H8, C4H10 और मीिेन के ष्टमश्रण के 

प्रभाव का अध्ययन ष्टकया है। ख) र्ाद के उष्ण और रासायष्टनक उपचार के र्ाद ष्टिओलाइि पर इन 

प्रािष्टमक मोष्टलब्डेनम ऑक्साइड/कार्ागइड प्रिाष्टियो ंको स्थिर करने के ष्टलए एक और रणनीष्टि प्रस्ताष्टवि 

है। यह मोष्टलब्डेनम अग्रदूिो ंके उष्ण उपचार और ऑिोरोस्िक α-MoO3 को हाइडर ोिमगल िोपोिैस्िक 

पररविगन के माध्यम से मेिासे्टर्ल हेक्सागोनल h-MoO3 में पररवष्टिगि करके प्राप्त ष्टकया िािा है। इस 

काम में, उते्प्ररक प्रष्टिष्टियािीलिा को र्दलने में ष्टिओलाइि के साि मोष्टलब्डेनम पूवगगामी परस्पर ष्टिया 

की भूष्टमका का अध्ययन ष्टकया िया है; ििा ग) उच्च एरोमैष्टिक (र्ेंिीन) चयनात्मकिा प्राप्त करने के 

ष्टलए HZSM-5 ष्टिओलाइि की ब्ोसें्टड अम्लिा को र्ोररक एष्टसड के साि प्रीिर ीि करके ष्टनयंष्टिि ष्टकया 

िािा है। इसके अलावा, HZSM-5 ष्टिओलाइि की ब्ोसें्टड अम्लिा को ष्टनयंष्टिि करने की यह नई 

रणनीष्टि सष्टिय Mo2C6 अणुओ ंके समूह और कार्गन िमाव को कम करके MDA में उते्प्ररक स्थिरिा 

को र्ढािी है। कुल ष्टमलाकर, इन अध्ययनो ंके माध्यम से, िोर् प्ररं्र् प्रष्टिष्टिया के सष्टिय उते्प्ररक साइिो ं

के आणष्टवक स्तर की अंिदृगष्टि प्रदान करिा है। र्हरहाल, यह अध्ययन MDA प्रष्टिया के ष्टलए उच्च 

उते्प्ररक स्थिरिा प्राप्त करने की क्षमिा के साि, वांष्टिि प्रष्टिष्टियािीलिा और उत्पाद चयनात्मकिा प्राप्त 

करने के ष्टलए मोष्टलब्डेनम अग्रदूि, कार्गराइष्टिंग स्थिष्टियो ंऔर ष्टिओलाइि उपचार की पसंद को ष्टनष्टदगि 

करके िकग संगि उते्प्ररक ष्टडिाइन के ष्टलए एक मागगदिगक उपकरण प्रदान करिा है। 
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